Author's response to reviews

Title: A human in vitro model system for investigating genome-wide host responses to SARS coronavirus infection

Authors:

Lisa F.P. Ng (ngfp@gis.a-star.edu.sg)
Martin L Hibberd (hibberdml@gis.a-star.edu.sg)
Eng-Eong Ooi (gptoee@sgh.com.sg)
Kin-Fai Tang (TANG_Kin_Fai@env.gov.sg)
Soek-Ying Neo (neosy3@gis.a-star.edu.sg)
Jenny Tan (tanh@gis.a-star.edu.sg)
Karuturi R. Krishna Murthy (karuturikm@gis.a-star.edu.sg)
Vinsensius B Vega (vegav@gis.a-star.edu.sg)
Jer-Ming Chia (jerm@gmx.net)
Edison T Liu (liue@gis.a-star.edu.sg)
Ee Chee Ren (renec@gis.a-star.edu.sg)

Version: 5  Date: 9 September 2004

Author's response to reviews:

6th Sep 2004

Dear BioMed Central Editorial Team,

Thank you for your reply on the resubmitted manuscript. We noted all your questions and comments, and have addressed them accordingly. The details of our response are given below:

MS: 8874017623968129 - A human in vitro model system for investigating genome-wide host response to SARS coronavirus infection

Answers to Editors' questions

Formatting changes requested

1. "Affiliations- Please remove the word 'Address' from the affiliation list"
   Ans: Changed accordingly.

2. "Figures- The figures should be cropped to minimise white space around the image. Our online figure cropping guide may be of some help, and can be viewed here:
   http://www.biomedcentral.com/info/ifora/cropping"
   Ans: All figures (Figures 1, 2, 3 and the supplementary figure) have been cropped to minimize the white spaces around the image.

Thank you so much in advance and we look forward to hearing from you soon.

Warmest regards,

Lisa F. P. Ng,
Martin L. Hibberd
Ee Chee Ren